Author’s response to reviews

Title: Fertility Sparing Surgery in advanced stage malignant ovarian germ cell tumor: A case report.

Authors:

Montassar Ghalleb (montaghalleb@gmail.com)
Hatem Bouzaiene (hatembouzaiene@yahoo.fr)
skander slim (slim.skander@gmail.com)
Achraf Hadiji (achrafhadiji@yahoo.fr)
Monia Hechiche (moniaHechiche@rns.tn)
Jamel Ben Hassouna (jamelbenhassouna@gmail.com)
Khaled Rahal (khaledrahal@yahoo.com)

Version: 2 Date: 16 Nov 2017

Author’s response to reviews:

Dear reviewer,

Please find below our responses to your comments point by point.

1) When initially examined, the patient presented acute abdominal pain, abdominal tenderness. A mobile tender mass was also palpated in the vaginal touch. No other signs were observed (neurological or other).

2) Unfortunately, there are no pictures available of the initial tumor. The patient was initially taken care of in another department sixteen years ago. Only the histological report was available to our team.
3) The six cures of BEP were well tolerated.

Thank you very much for your time and efforts.

Kind regards,

Ghalleb Montassar